Author's response to reviews

Title: Acute Myopathy secondary to oral steroid therapy in a 49 year old man: A case report.

Authors:

Muhammad A Khan (aleekhan@gmail.com)
Eric Larson (larson.eric@usd.edu)

Version: 2 Date: 14 October 2010

Author's response to reviews:

October 12, 2010.

Editor-in-Chief, Journal of Medical Case Reports,
JMCR
BioMed Central Ltd
Floor 6, 236 Gray's Inn Road
London WC1X 8HL
United Kingdom

Dear Editor,

On behalf of all the authors, I would like to thank you for the suggestions. We feel that this would only improve upon the content of the report. Necessary changes have been made and a revised case report titled “Acute Myopathy secondary to oral steroid therapy in a 49 year old man: a case report” is uploaded.

Patient suffered from gastroesophageal reflux disease prior to the reported illness. Patient continues to complain of foot pain consistent with plantar fascitis. He has been managing this symptomatically with anti-inflammatory medications.

Changes have been highlighted for easy viewing. We will continue to make amendments if needed necessary. Please let us know if further changes need to be made or if we need to clarify any existing point.

Sincerely,

Muhammad Ali Khan, MD
Resident department of Internal Medicine,
Sanford School of Medicine,
University of South Dakota.
Health Science Center,